Author’s response to reviews

Title: Maternal plasma levels of oxytocin during physiological childbirth – a systematic review with implications for uterine contractions and central actions of oxytocin

Authors:

Kerstin Uvnäs Moberg (k_uvnas_moberg@hotmail.com)

Marie Berg (marie.berg@fhs.gu.se)

Anette Ekström (anette.ekstrom@his.se)

Sarah Buckley (drsarahjbuckley@gmail.com)

Soo Downe (SDowne@uclan.ac.uk)

Eleni Hadjigeorgiou (eleni.hadjigeorgiou@cut.ac.cy)

Bogumila Kielbratowska (bogumila.kielbratowska@gumed.edu.pl)

Alicja Kotlowska (kotlowska@gumed.edu.pl)

Luise Lengler (luise.lengler@t-online.de)

Fatima Leon-Larios (fatimaleon@us.es)

Bengt Lindström (bengtblind@hotmail.com)

Claudia Meier Magistretti (claudia.meiermagistretti@hslu.ch)

Zada Pajalic (zada.pajalic@hioa.no)

Anna Dencker (anna.dencker@gu.se)

Version: 5 Date: 31 May 2019

Author’s response to reviews:

Dear BMC Pregnancy and Childbirth,

Thank you for your positive message and for your valuable comments regarding our ms: PRCH-D-18-00653R4 Maternal plasma levels of oxytocin during physiological childbirth – a systematic review with implications for uterine contractions and central actions of oxytocin
The articles you suggest are very interesting, but they do not follow the inclusion criteria as they concern pregnancy and therefore we did not include them in this review.

On the other we have included them in another review in regarding mood mental states, attachment etc and oxytocin levels during pregnancy, labour and breastfeeding.

Best regards Anette Ekström-Bergström & Kerstin Uvnäs-Mpberg

Anette Ekström-Bergström

RNM, PhD, Professor in Nursing

University Väst

Department of Health Sciences

Gustav Melins gata 2

461 32 Trollhättan, Sverige

+46 520 22 39 72

+4673 901 34 41

anette.ekstrom-bergstrom@hv.se

https://www.researchgate.net/profile/Anette_Ekstroem-Bergstroem